Author’s response to reviews

Title: Morbidity and mortality in a population of patients affected by heart failure and chronic obstructive pulmonary disease: an observational study

Authors:

Eugenio Cosentino (submission@polistudium.it)
Matteo Landolfo (m.landolfo.doc@gmail.com)
Crescenzio Bentivenga (crescenzio.bentivenga@asops.bo.it)
Luca Spinardi (luca.spinardi@aosp.bo.it)
Daniela Esposti (daniela.deglasposti@aosp.bo.it)
Arrigo Cicero (arrigo.cicero@unibo.it)
Rinaldo Miceli (rinaldo.miceli@gmail.com)
Virna Bui (virna.bui2@unibo.it)
Emanuela Berardi (emanuela.berardi@gmail.com)
Claudio Borghi (claudio.borghi@unibo.it)

Version: 2 Date: 16 Nov 2018

Author’s response to reviews:

To whom it may concern,

Many thanks for taking the time to review our manuscript.

We appreciate your input and have now revised our manuscript "Morbidity and mortality in a population of patients affected by heart failure and chronic obstructive pulmonary disease: an observational study" (BCAR-D-18-00521R1) as per your suggestions:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated. Done
2. In the “Ethics Approval and Consent to Participate” section please include the full name of
the ethics committee (and the institute to which it belongs to) that approved the study and
the committee’s reference number if appropriate. Done

3. Please confirm in the section 'Ethics Approval and Consent to Participate' whether informed
consent, written or verbal, was obtained from all participants and clearly state this in your
manuscript. If verbal, please state the reason and whether the ethics committee approved this
procedure. If the need for consent was waived by an IRB or is deemed unnecessary
according to national regulations, please clearly state this, including the name of the IRB or
a reference to the relevant legislation. Done

4. Please represent individual authors' contributions using their full initials, in the Authors’
Contributions section. At the moment this section is blank. Please ensure all authors meet
the criteria for authorship in order to avoid further delays:

An 'author' is generally considered to be someone who has made substantive intellectual
contributions to a published study. According to the ICMJE guidelines, to qualify as an author
one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis
and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual
content; AND

c) given final approval of the version to be published. Each author should have participated
sufficiently in the work to take public responsibility for appropriate portions of the content;
AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the
accuracy or integrity of any part of the work are appropriately investigated and resolved.

Done

Anyone listed as an author must be included in this section. If you choose to change your author
list you will need to fill out a change in authorship form and send it by email to the Editorial
office to be approved by the Editor. The form can be found here:
https://www.biomedcentral.com/getpublished/editorial-policies#authorship.
Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section. Done

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Done

Many thanks,

Eugenio Roberto Cosentino (on behalf of all authors)